

RHEUMATOLOGY

Clinical science

Lung function and skin fibrosis changes as predictors of survival in SSc-associated interstitial lung disease: a **EUSTAR study**

Vincent Sobanski (1,2,*, Jeska de Vries-Bouwstra³, Anna-Maria Hoffmann-Vold (1,5,*, Dörte Huscher⁶, Margarida Alves⁷, Marco Matucci-Cerinic^{8,9}, Gabriela Riemekasten ^{[0]10}, Mengtao Li 11,12, László Cziriák¹³, Otylia Kowal-Bielecka¹⁴, Yannick Allanore 15, Nils Schoof¹⁶, Oliver Distler ⁵, on behalf of the EUSTAR Collaborators

¹Service de Médecine Interne, U1286 - INFINITE—Institute for Translational Research in Inflammation CHU Lille, Univ. Lille, Inserm, Lille, France

²Institut Universitaire de France (IUF), Paris, France

³Department of Rheumatology, Leiden University Medical Center, Leiden, The Netherlands

⁴Department of Rheumatology, Oslo University Hospital, Oslo, Norway

⁵Department of Rheumatology, University Hospital Zurich, University of Zurich, Zurich, Switzerland

⁶Institute of Biometry and Clinical Epidemiology, and Berlin Institute of Health, Charité—Universitätsmedizin Berlin, Berlin, Germany

⁷TA Inflammation Med, Boehringer Ingelheim International GmbH, Ingelheim am Rhein, Germany

⁸Unit of Immunology, Rheumatology, Allergy and Rare Diseases (UnIRAR), and Inflammation, Fibrosis and Aging Initiative (INFLAGE), IRCCS San Raffaele Scientific Institute, Rome, Italy

⁹Vita-Salute San Raffaele University, Milan, Italy

¹⁰Department of Rheumatology and Clinical Immunology, University of Lübeck, Lübeck, Germany

¹¹Department of Rheumatology and Clinical Immunology, Peking Union Medical College Hospital, Chinese Academy of Medical Sciences and Peking Union Medical College, Beijing, China

¹²Ministry of Education, Key Laboratory of Rheumatology and Clinical Immunology, Beijing, China

¹³Department of Rheumatology and Immunology, Medical School, University of Pecs, Pecs, Hungary

¹⁴Department of Rheumatology and Internal Medicine, Medical University of Bialystok, Bialystok, Poland
¹⁵Rheumatology Department, Cochin Hospital APHP, INSERM U1016, Université Paris Cité, Paris, France

¹⁶Integrated Evidence Generation TA WHC, Bayer AG, Berlin, Germany

*Correspondence to: Vincent Sobanski, Service de Médecine Interne, CHU Lille, Hôpital Claude Huriez, rue Michel Polonowski, F-59000 Lille, France; Institut Universitaire de France (IUF), 103 boulevard Saint-Michel, 75005 Paris, France. E-mail: vincent.sobanski@univ-lille.fr

[†]See Supplementary Material, available at *Rheumatology* online for a list of the EUSTAR Collaborators.

Abstract

Objectives: This study assessed how changes in lung function, skin fibrosis and digital ulceration (DU) burden predict mortality in patients with SSc-associated interstitial lung disease (SSc-ILD), the leading cause of death in SSc.

Methods: Adult SSc-ILD patients from the European Scleroderma Trials and Research (EUSTAR) database enrolled since January 2009 with a date of diagnosis, a follow-up visit for change evaluation within 12 months plus a further visit or mortality information were eligible. Twelvemonth changes in lung function (per cent predicted forced vital capacity [FVC%pred] and diffusing capacity of the lungs for carbon monoxide [DL_{co}%pred]), modified Rodnan skin score (mRSS) and change in DU burden were assessed for associations with survival, using multivariable Cox regression analyses adjusted for age, sex, smoking status and immunosuppressive therapy.

Results: Of 893 SSc-ILD patients included, 94 (10.5%) died over a mean follow-up of 39.0 ± 23.9 months. Absolute deterioration in FVC >10% pred within 12 months (n = 78/638 evaluable) was predictive for decreased survival (hazard ratio [HR] 3.81; 95% CI 1.67–8.66), as were composite measures combining (i) >10% FVC decline or mRSS worsening (HR 2.82; 95% Cl 1.43–5.56) and (ii) FVC decline ≥10% or 5–9% with DL_{CO} decline ≥15% (HR 3.42; 95% CI 1.68–7.00), but not changes in DL_{CO}, mRSS or DU burden alone.

Conclusions: Changes in lung function and skin fibrosis within 12 months should be considered when evaluating risk of mortality. The effect of pharmacological treatments aiming at stabilization of these variables should be evaluated prospectively in clinical trials.

Keywords: SSc-ILD, systemic sclerosis, interstitial lung disease, mortality, prognostic stratification.

Received: 10 December 2024. Accepted: 1 May 2025

© The Author(s) 2025. Published by Oxford University Press on behalf of the British Society for Rheumatology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Rheumatology key messages

- SSc progression, assessed by 12-month changes in lung function and skin fibrosis, can predict mortality.
- Regular assessments of lung function and mRSS are crucial for understanding SSc(-ILD) prognosis and management.
- Identification of SSc-ILD patients at higher risk of progression should be of priority for clinicians.

Introduction

Interstitial lung disease (ILD) is a leading cause of morbidity and mortality in SSc, affecting >50% of patients [1], and is responsible for 15-33% of deaths in SSc [2-4]. Up to onethird of patients will develop progressive ILD within 12 months, leading to more extensive lung changes and worsening of fibrosis [1, 5]. Patients with ILD may develop loss of vital capacity during the first few years after SSc diagnosis [6], highlighting the need for early and regular screening [1]. Once ILD is identified, decisions regarding management can be challenging, given that some patients may not progress, and it may be difficult to identify progression at an early stage [7]. Age, male sex, history of smoking, low forced vital capacity (FVC), low diffusing capacity of the lungs for carbon monoxide (DL_{CO}) and extent of lung fibrosis on highresolution CT (HRCT) at baseline are associated with mortality in patients with SSc-associated ILD (SSc-ILD) [8-10].

Recently, agents such as pirfenidone, rituximab and tocilizumab have been investigated in clinical trials, alone or in combination with traditional immunosuppressants [11–13]. Patient selection for treatment is often based on the risk of adverse outcomes, including mortality. Thus it is important to be able to identify patients who are at highest risk of death [14].

Previous studies have suggested that short-term pulmonary function test (PFT) changes could be predictive of mortality, and composite indexes might add value [8, 15, 16]. In 162 patients of the Royal Brompton Hospital (London, UK) cohort, Goh et al. [17] determined that an FVC/DL_{CO} composite end point over a 12-month period (absolute FVC decline $\geq 10\%$ predicted or 5–9% with absolute DL_{CO} decline $\geq 15\%$ predicted) was predictive for mortality. In a *post hoc* analysis of Scleroderma Lung Studies (SLS) I and II [16], a decline in FVC and DL_{CO} over 2 years was a better predictor of mortality than baseline FVC and DL_{CO}, with composite measures better predictors than changes in individual PFT scores. Changes in digital ulceration (DU) burden have also been associated with survival in a previous study [18]. Composite indices are increasingly used in clinical trials [19]. The current study therefore assessed the predictive ability of changes in lung function, skin fibrosis and DU burden, and combinations of these, for mortality in a large European cohort of patients with SSc-ILD. A plain language summary of this article can be found in Supplementary Data S1, available at Rheumatology online.

Methods

Cohort selection

This longitudinal cohort study was based on data from patients with SSc-ILD included in the European Scleroderma Trials and Research (EUSTAR) database [20], with visits recorded between 1 January 2009 and 3 May 2018. The eligibility criteria were: age \geq 18 years; SSc according to the 1980 ACR or 2013 ACR/EULAR classification criteria [21, 22] and ILD according

to HRCT and/or X-ray imaging and/or confirmed ILD diagnosis reported by the treating physician (Fig. 1).

The baseline visit was the date of enrolment into the database for prevalent cases (existing ILD for >6 months at enrolment into the database or unknown date of diagnosis) or the date of ILD diagnosis for incident cases (ILD diagnosis \leq 6 months at inclusion or a new ILD diagnosis during followup). Follow-up was from the baseline visit until the last available documentation, withdrawal from the database or death. Patients were excluded if (i) they had no follow-up visit for change evaluation within 12 months after the baseline visit, or (ii) they had a follow-up visit for change evaluation after the baseline visit but no further visit and no information regarding mortality (Fig. 1). Pulmonary hypertension (PH), defined by right heart catheterization, could not be included because of missing data. The variable 'PH suspected by echocardiogram' was instead used, defined as increased systolic pulmonary arterial pressure or indirect signs of PH based on the clinician's judgement. Treatment data were available for the selected cohort, but the indications for these therapies were unavailable. At the time of database extraction, data on pirfenidone or nintedanib were not registered.

Outcomes

The cohort was analysed for associations with all-cause mortality of the following parameters (Table 1): (i) absolute changes from the baseline visit up to 12 months in per cent predicted values for FVC (FVC%pred) and DL_{CO} (DL_{CO} %pred); (ii) absolute and relative changes from baseline to 12 months in modified Rodnan skin score (mRSS) and (iii) change in DU burden (development or persistence of DU vs remaining DU free and healed DU). Some parameters used in SSc clinical trials were not included due to rarity (history of scleroderma renal crisis ~3%; not meaningful for multivariable analysis) or missing values (PH defined by right heart catheterization) in the EUSTAR cohort [23].

In addition, two composite disease progression definitions were analysed for association with survival: (i) lung progression defined as absolute decline in FVC $\geq 10\%$ pred or absolute decline in FVC 5–9% pred with decline in DL_{CO} $\geq 15\%$ pred [17], and (ii) lung or skin progression defined as absolute decline in FVC >10\% pred or increase of mRSS >5 points and by >25% [24].

Ethics

This study was conducted in accordance with the Declaration of Helsinki and was approved by the local ethical committees of the participating EUSTAR centres. All patients provided written informed consent for their data to be used for research purposes as required by the local ethics committees for this study.

Statistical analyses

Comparisons were performed using Student's t test or Mann-Whitney U test for continuous variables, and Fisher's exact test or Chi-squared test for categorical variables.



Figure 1. Flow chart of the patients through the study. EUSTAR, European Scleroderma Trials and Research; HRCT, high-resolution CT; ILD, interstitial lung disease; SSc-ILD, SSc-associated interstitial lung disease

Survival time was defined as the duration between the visit of change evaluation and the last visit with survival status, censored after 5 years of observation to maintain reasonable case numbers. The effect of changes in lung function, skin and DU burden on survival was evaluated using univariable analyses (Kaplan-Meier) for group comparisons and multivariable analyses (Cox proportional hazards models) adjusted for age, sex, smoking status and immunosuppressive therapy, as these parameters were judged to be clinically important for modifying major organ involvement. In multivariable analysis, the duration of ILD at enrolment as an additional confounder was considered for prevalent patients. For patients with an unknown date of ILD diagnosis, the duration of non-RP symptoms was used. Due to missing data on PH, as defined by right heart catheterization in the EUSTAR database, PH could not be included as a covariate in this analysis. Instead, PH suspected by echocardiogram was used as a covariate in a second multivariable analysis. Confounders with >20% missing values were omitted from multivariable analysis. P-values <0.05 were considered statistically significant. All analyses were conducted using IBM SPSS Statistics version 24.0 (IBM Corp, Armonk, NY, USA).

Results

Patient disposition

Of 14 804 patients screened, 3145 eligible patients with SSc-ILD were identified; 2013 had no follow-up visit within 12 months after the baseline visit (including 87 who died without a follow-up visit) and 239 had no follow-up after this evaluation. Therefore, the study population consisted of 893 patients, including 660 prevalent cases (73.9%) and 233 incident cases (26.1%), all of whom were included in the analysis (Fig. 1). Of these, 94 (10.5%) died over a mean follow-up of 39.0 ± 23.9 months. Of the 94 deaths, 81 (86.2%) were declared as SSc-related. Demographic and baseline characteristics of included and excluded patients are shown in Table 1. There were no significant differences between the study population and excluded patients with regard to sex, disease duration, proportion of dcSSc, joint and muscular involvement, CRP elevation and baseline PFT results. Differences were observed in other parameters, some of which may have been related to inclusion/exclusion criteria: patients in the excluded group were older, had a shorter follow-up, a higher prevalence of PH and ACA, and were receiving less immunosuppressive therapy than the study cohort. Detailed treatment data for patients are included in Supplementary Table S1, available at Rheumatology online.

Changes during the first 12 months

The median time from baseline to the visit of change evaluation ranged between 8.5 and 9 months in groups according to the parameters in Table 2 and did not differ by >1 month between categories within each parameter group, with the exception of DU, for which there was a difference of 2 months.

At 12 months, 363 patients (56.9%) had no decline in FVC, 197 (30.9%) had a decline of >0-10% pred, and 78 (12.2%) had a decline in FVC >10% pred. The percentage

Table 1. Demographic and baseline characteristics of included and excluded SSc-ILD patients in the EUSTAR database

Variables	n	Included <i>n</i> patients (<i>N</i> =893) ^a		Excluded patients $(N = 2252)^a$	
Demographics, n (%)					
Age (years), mean \pm S.D.	893	55.6 ± 12.8	2252	56.9 ± 13.4	
Male sex	893	172 (19.3)	2252	422 (18.7)	
Disease duration (years), ^b median (IQR)	775	5.1 (2.5-10.4)	1939	5.6 (2.3-11.6)	
Follow-up (months), mean \pm S.D.	893	39.0 ± 23.9	1212	34.6 ± 25.7	
Extra-pulmonary characteristics, <i>n</i> (%)					
dcSSc	849	370 (43.6)	2149	931 (43.3)	
mRSS (units), mean \pm S.D.	790	11.0 ± 9.2	2052	10.6 ± 9.1	
2001 EUSTAR SSc activity score >3	892	201 (22.5)	2252	540 (24.0)	
DU (past or active)	665	411 (61.8)	1574	971 (61.7)	
Joint synovitis	874	133 (15.2)	2199	350 (15.9)	
Muscle weakness	873	181 (20.7)	2193	473 (21.6)	
Pulmonary hypertension suspected by echocardiography	726	142 (19.6)	1861	432 (23.2)	
LVEF, mean ± S.D.	662	62.0 ± 6.3	1669	61.7 ± 7.2	
Renal crisis	884	12 (1.4)	2236	53 (2.4)	
ACA positive	811	139 (17.1)	1970	430 (21.8)	
Anti-topo I positive	828	439 (53.0)	2035	1007 (49.5)	
CRP elevation	831	252 (30.3)	2031	632 (31.1)	
Immunosuppressive therapy ^c	893	469 (52.5)	2251	945 (42.0)	
Lung-associated characteristics, n (%)					
Prevalent SSc-ILD	893	660 (73.9)	2252	1645 (73.0)	
Incident SSc-ILD	893	233 (26.1)	2252	607 (26.0)	
FVC (%pred), mean ± S.D.	754	86.4 ± 21.1	1843	85.9 ± 21.9	
DL_{CO} (%pred), mean ± S.D.	708	60.6 ± 20.7	1729	60.7 ± 19.0	
Dyspnoea NYHA ≥2	854	493 (57.7)	2072	1209 (58.3)	

^a Patients may not have had data for every variable, and results for each variable are expressed as a percentage of those with available data. For each group, *n* indicates the number of patients with available data.

^bCalculated as difference between the date of the baseline visit and the date of the first non-RP symptom of the disease as reported by the patients. ^cDefined as treatment with prednisone in doses >10 mg/day or any immunosuppressant (CYC, SSZ, MTX, LEF, AZA, MMF, ciclosporin A, D-Pen, ituximab, imatinib, anti-TNF-α, tocilizumab, abatacent or other biologic therapies) at baseline.

rituximab, imatinib, anti-TNF-*a*, tocilizumab, abatacept or other biologic therapies) at baseline. %pred, per cent predicted; DL_{CO}, diffusing capacity of the lungs for carbon monoxide; DU, digital ulceration; EUSTAR, European Scleroderma Trials and Research; FVC, forced vital capacity; IQR, interquartile range; LVEF, left ventricular ejection fraction; mRSS, modified Rodnan skin score; NYHA, New York Heart Association; SSc-ILD, SSc-associated interstitial lung disease.

of patients with changes in the disease parameters analysed at 12 months is shown in Table 2.

Survival analyses

In the whole cohort, 94 (10.5%) patients died during the follow-up period (mean 39.0 ± 23.9 months). Kaplan–Meier analyses identified significantly shorter survival in patients with progression of certain parameters within 12 months of the baseline visit: (i) those with a deterioration in FVC >10% pred (P < 0.001) (n = 78) (Fig. 2A); (ii) those with a decline in FVC ≥10% pred or a decline in FVC of 5–9% pred with a decline in DL_{CO} ≥15% pred (n = 96) (P < 0.001) (Fig. 2B) and (iii) those with a decline in FVC >10% pred or a worsening of mRSS >5 points and >25% (P = 0.002) (n = 138) (Fig. 2C). No associations with survival were found for isolated changes in DL_{CO}% pred, mRSS or DU burden.

Multivariable Cox proportional hazard models with adjustment for age, sex, smoking status, duration of ILD and immunosuppressive therapy confirmed the findings of the Kaplan–Meier analyses, with hazard ratios (HRs) for mortality associated with changes in parameters as follows: (i) patients with a decline in FVC >10% pred (HR 3.81; 95% CI 1.67–8.66); (ii) patients with a decline in FVC \geq 10% pred or a decline in FVC 5–9% pred with a decline in DL_{CO} \geq 15% pred (HR 3.42; 95% CI 1.68–7.00) and (iii) patients with a decline in FVC >10% pred or an increase in mRSS >5 points and >25% (HR 2.82; 95% CI 1.43–5.56) (Table 3).

Because of missing data regarding PH defined by right heart catheterization in the EUSTAR database, this variable was not included in the analysis. However, a second multivariable analysis was performed as a sensitivity analysis, including the variable PH suspected by echocardiogram as a covariate. Results were similar although non-significant, with HRs for mortality associated with changes within 12 months of the baseline visit as follows: (i) HR 2.32 (95% CI 0.86-6.25) for decline in FVC >10% pred (P = 0.097); (ii) HR 2.08 (95% CI 0.86–5.06) for decline in FVC \geq 10% pred or decline in FVC 5-9% pred with decline in $DL_{CO} \ge 15\%$ pred (P=0.105) and (iii) 2.30 (95% CI 0.97-5.47) for decline in FVC >10% pred or worsening in mRSS >5 points and >25% (P=0.059) (Supplementary Table S2, available at Rheumatology online). When excluding patients with suspected PH, the HRs for mortality were as follows: (i) HR 2.34 (95% CI 0.72-7.56) for decline in FVC >10% pred (P = 0.155); (ii) HR 2.75 (95% CI 0.98–7.77) for decline in FVC $\geq 10\%$ pred or decline in FVC 5–9% pred with decline in $DL_{CO} \ge 15\%$ pred (P = 0.056) and (iii) 2.88 (95% CI 1.18-7.03) for decline in FVC >10% pred or worsening in mRSS >5 points and >25% (P=0.020) (Supplementary Table S3, available at Rheumatology online).

Discussion

These data indicate that changes in pulmonary function and skin fibrosis over 12 months were predictive for mortality in patients with SSc-ILD and should be considered in the prognostic stratification of these patients. Table 2. Disease parameters and categories of change assessed for prediction of mortality

Parameter	Categories	Patients meeting the endpoints
Absolute change from baseline in FVC	Increase or no decline in FVC	• 363 (56.9%)
	• Decline in FVC >0–10%pred	• 197 (30.9%)
	• Decline in FVC >10% pred	• 78 (12.2%)
Absolute change from baseline in DL_{CO}	• Increase or no decline in DL _{CO}	• 318 (53.6%)
	• Decline in DL _{CO} >0–10%pred	• 183 (30.9%)
	• Decline in DL _{CO} >10-15%pred	• 39 (6.6%)
	• Decline in DL _{CO} >15% pred	• 53 (8.9%)
Change from baseline in mRSS [25]	• Improved or stable mRSS	• 509 (66.9%)
	 Worsening in mRSS ≤5 points and/or ≤25% 	• 183 (24.0%)
	• Worsening in mRSS >5 points and >25%	• 69 (9.1%)
Change in DU burden	• Remaining DU free or healed DU	• 612 (72.6%)
	• Development or persistence of DU	• 231 (27.4%)
Lung progression: composite of FVC and DL _{CO} [17]	• Increased/stable FVC or decline $<10\%$ pred, with decline in DL _{CO} $<15\%$ pred	• 539 (84.9%)
	• Decline in FVC ≥10%pred or 5–9%pred, with decline in DL _{CO} ≥15%pred	• 96 (15.1%)
Lung and skin progression: composite of FVC and mRSS [24]	 Increased/stable FVC or decline ≤10% pred, with improved/stable mRSS or worsening by ≤5 points or ≤25% 	• 454 (76.7%)
	 Decline in FVC >10%pred or worsening mRSS by >5 points and >25% 	• 138 (23.3%)

%pred, per cent predicted; DL_{CO}, diffusing capacity of the lungs for carbon monoxide; DU, digital ulceration; FVC, forced vital capacity; mRSS, modified Rodnan skin score.



Figure 2. Overall survival according to (A) change in FVC, (B) change in FVC and/or DL_{CO} and (C) change in FVC and/or mRSS. All changes within 12 months of the baseline visit. Categories for FVC and DL_{CO} are based on %pred values. %pred, per cent predicted; DL_{CO} , diffusing capacity of the lungs for carbon monoxide; FVC, forced vital capacity; mRSS, modified Rodnan skin score

Given the high clinical heterogeneity of ILD in SSc, one of the main challenges that clinicians have to address when following SSc-ILD patients is to identify those at a higher risk of poor outcome [23]. PFTs offer a simple and non-invasive evaluation of lung involvement at initial diagnosis and during follow-up. However, PFTs do not discriminate between SSc patients with and without radiographic SSc-ILD [26]. Goh *et al.* developed an algorithm showing that extent of ILD >20% on HRCT, or lesser degrees of ILD combined with FVC <70\%, might predict mortality [9]. This has been replicated in a subsequent study [27]. Until recently, baseline FVC has been an unadjusted predictor of mortality in multiple

Table 3. Predictors of mortality in SSc-ILD in the EUSTAR database in uni- and multivariable Cox regression analyses

	Univariable analysis ^a		Multivariable analysis ^a	
	HR (95% CI)	P-value	HR (95% CI)	P-value
FVC % predicted change				
Increase or no decline	Reference		Reference	
Decline >0-10%	1.10 (0.58-2.10)	0.761	1.30 (0.60-2.84)	0.511
Decline >10%	3.06 (1.59-5.88)	0.001	3.81 (1.67-8.66)	0.001
DL_{CO} % predicted change				
Increase or no decline	Reference		Reference	
Decline >0-10%	0.94 (0.47-1.85)	0.847	0.66 (0.26-1.68)	0.386
Decline >10-15%	1.07 (0.33-3.55)	0.907	1.67 (0.48-5.80)	0.419
Decline >15%	0.96 (0.33-2.74)	0.934	0.33 (0.04-2.53)	0.283
mRSS change				
Decline or no increase	Reference		Reference	
Increase ≤ 5 points and/or $\leq 25\%$	0.68 (0.37-1.26)	0.219	0.93 (0.46-1.86)	0.827
Increase >5 points and $>25\%$	1.17 (0.53-2.58)	0.706	2.06 (0.89-4.73)	0.090
DU				
DU free or healed DU	Reference		Reference	
New or persisting DU	0.97 (0.60-1.56)	0.888	0.75 (0.39-1.44)	0.392
Composite FVC and DL _{CO} change			· · · ·	
Increased/stable FVC or $<10\%$, with DL _{co}	Reference		Reference	
decline <15%				
FVC decline $\geq 10\%$ or 5–9%, with DL _{CO} decline $\geq 15\%$	2.69 (1.50-4.83)	0.001	3.42 (1.68-7.00)	0.001
Composite FVC and mRSS change	, , , , , , , , , , , , , , , , , , ,			
FVC increased/stable/or decline $\leq 10\%$, with mRSS	Reference		Reference	
improved/stable or worsened by ≤ 5 points or $\leq 25\%$				
FVC decline >10%, or mRSS worsened by >5 points and >25%	1.99 (1.13-3.52)	0.018	2.82 (1.43-5.56)	0.003

^a Models adjusted for age, sex, tobacco use and immunosuppressive therapy (both analyses) plus duration of ILD (multivariable analysis only). DL_{CO}, diffusing capacity of the lungs for carbon monoxide; DU, digital ulceration; EUSTAR, European Scleroderma Trials and Research; FVC, forced vital capacity; HR, hazard ratio; ILD, interstitial lung disease; mRSS, modified Rodnan Skin Score; SSc-ILD, SSc-associated interstitial lung disease. Bold text indicates significance: P < 0.05.

studies, but was not identified as an independent predictor in any multivariable analyses [8]. DL_{CO} has been identified in multiple studies as a significant predictor of mortality in both unadjusted and adjusted analyses, probably because DL_{CO} is a marker of ILD and PH, two major contributors to mortality [8]. The widespread use of FVC may underestimate the potential for other predictors of SSc-ILD progression. Failing to fully identify treatment effects in SSc-ILD randomized controlled trials, which was described in a recent systematic review, highlights the need to pursue efforts to identify the best sole or composite PFT surrogate marker for SSc-ILD [28].

FVC is one of the most reproducible physiological assessments of lung disease, although definitions of progression used in idiopathic pulmonary fibrosis (IPF) and SSc trials, such as a $\geq 10\%$ relative decline in FVC, may not reflect the natural history of decline in SSc-ILD [29]. A recent study found that the FVC in SLS I and II, measured at specialist centres in the USA, had acceptable test-retest reliability and that small changes in FVC (3-5.3%) correlated with patientreported outcomes and findings on HRCT, defining these as minimal clinically important differences [30]. Following a consensus exercise in 2015, the Outcome Measures in Rheumatology Connective Tissue Disease (CTD)-associated ILD working group proposed a clinically meaningful outcome in clinical trials of CTD-associated ILD (≥10% relative decline in FVC or $a \ge 5\%$ to <10% relative decline in FVC with $\geq 15\%$ relative decline in DL_{CO}) [31]. Data-driven approaches to validate these results in different cohorts and randomized controlled trials are needed. In this study, an absolute decline in FVC $\geq 10\%$ or 5–10% with a decline in $DL_{CO} \ge 15\%$ within 12 months of the baseline visit was associated with mortality, independent of age, sex, smoking status and immunosuppressive therapy. This is consistent with a recent study by Goh et al. that examined the prognostic significance of PFTs at 1 and 2 years against 15-year survival in patients with SSc-ILD, which found a composite measure to be the best predictor for mortality [17]. One-year PFT changes were predictive only in patients with extensive lung disease. In those with less fibrosis, composite indices combining smaller declines in FVC (<10%) with DL_{CO} decline $(\geq 15\%)$ at 2 years were more predictive for mortality. The extent of lung fibrosis on HRCT has only been recorded in the EUSTAR database since 2013, and therefore this item could not be evaluated. A recent post hoc analysis of SLS I and II also showed that decline in FVC and DL_{CO} over 2 years was a better predictor of mortality than baseline FVC and DL_{CO}, with composite measures better predictors than changes in individual PFT scores [16]. Together, these results suggest that short-term variations in surrogate measures of SSc-ILD progression might have important effects on longterm outcomes [16].

The main caveats to using FVC and/or DL_{CO} decline to define SSc-ILD worsening are that the reported time to decline is highly dependent on the intervals between PFTs (likely to be influenced by disease severity) and that FVC or DL_{CO} may vary during follow-up, so that patients fulfilling the criteria of 'significant decline' at one visit may not fulfil them at the following visit. An analysis of SSc-ILD progression in the EUSTAR database showed a highly variable and heterogeneous disease course [32]. Due to a lack of standardization, the formula for calculating FVC%pred and DL_{CO} %pred may also vary between centres. Using a linear mixed model that integrates all serial PFTs in SSc-ILD patients, Le Gouellec *et al.* found that during follow-up of 75 patients, FVC was

stable while DL_{CO} significantly decreased (-1.5 ± 0.3%/year) [33]. Presence or history of DU and presence of PH at baseline or during follow-up were associated with a faster decline of DL_{CO} over time. Some studies have suggested that FVC decline is most rapid earlier in the course of SSc-ILD and that ILD might be more stable after the first 4 years following ILD diagnosis [34, 35]. However, one study has suggested that this impression could be an artefact secondary to survival bias. Indeed, a plateau in the progression of FVC was apparent in the full cohort analysis but disappeared with stratification into prognostic subgroups to account for survival bias. Patients had distinct patterns of progression that remained relatively consistent during long-term follow-up. They further showed that recent change in FVC could not predict future change in FVC within shorter follow-up intervals; however, patients with a decline in DL_{CO} tended to have a continuous decrease in the subsequent year [15]. A EUSTAR study, which focused on long-term progression patterns in patients with SSc-ILD by assessing FVC decline through the 12 ± 3-month trial and in the 5-year follow-up period, showed that FVC had heterogeneous and variable trajectories. The proportion of patients with SSc-ILD who experienced FVC decline of $\geq 5\%$ during the initial 12 ± 3 -month period was 27%, and in each 12-month period over the mean 5-year follow-up, 23–27% of patients experienced progression [32].

The high heterogeneity in SSc does not allow a single treatment approach to be identified and also affects enrolment of patients in clinical trials [27]. European evidence-based consensus statements have provided several ways of assessing progression (in treated or untreated patients): changes in PFTs (FVC and DL_{CO} absolute values or FVC decline); changes in extent of fibrosis or pattern on HRCT; changes in exercise-induced oxygen desaturation and worsening of clinical symptoms [36]. Several predictive models using demographic and clinical data have been developed to assess prognosis in patients with ILD. The GAP (sex, age and lung physiology [i.e. FVC and DL_{CO}]) model was originally developed to assess 1-, 3- and 5-year mortality in IPF but has been applied and validated in patients with CTD, including SSc [37]. The SADL (smoking history, age, DL_{CO}) model has been validated specifically in SSc patients to predict all-cause mortality in SSc-ILD [10]. The SPAR (SpO₂, arthritis) model utilizes optimal cut-offs for desaturation of oxygen with 6min walk testing in combination with the presence of arthritis, which were both identified as independent predictors for ILD progression in patients with mild SSc-ILD [38]. In the current study, although change in mRSS alone was not associated with survival, unlike in a recent EUSTAR study in unselected patients with dcSSc [39], the composite of FVC decline 5-9% or mRSS change predicted mortality, but with a lower HR than FVC decline >10%. An explanation could be that progressive skin fibrosis, as assessed by mRSS, has been associated with lung function decline, assessed by changes in FVC, and worse survival during follow-up [39].

Immunosuppressive therapies remain the mainstay of treatment for SSc-ILD [40], but the drug armamentarium is still relatively limited for this severe and life-threatening complication of SSc. Numerous targeted therapies are being evaluated [41]. Studies with nintedanib (a tyrosine kinase inhibitor), tocilizumab (an IL-6 receptor inhibitor) and rituximab (an anti-CD20 antibody) have shown alleviation of the annual rate of decline in FVC among patients with SSc-ILD [11, 12, 42]. Nintedanib is approved for patients with SSc-ILD worldwide, tocilizumab is approved for patients with SSc-ILD in the United States and rituximab is approved for patients with SSc in Japan [43]. Riociguat, a soluble guanylate cyclase stimulator that has been evaluated in patients with early dcSSc (RISE-SSc), has shown some potential efficacy signals in this population [44, 45]. Additionally, many drugs, either approved or in development for SSc and progressive pulmonary fibrosis (including SSc-ILD), target additional, different mechanisms [13, 46, 47]. For example, pirfenidone, an antifibrotic agent that is approved for use in IPF and evaluated in guidelines for progressive pulmonary fibrosis, has been assessed in patients with SSc-ILD [13], with a recent American Thoracic Society guideline recommendation for additional research into its efficacy and safety in this population, either alone or in combination with MMF [40]. The phosphodiesterase-4 inhibitor nerandomilast has shown antifibrotic and anti-inflammatory properties in preclinical studies [46] and stabilized lung function over 12 weeks in patients with IPF [47]. Nerandomilast is currently being investigated, along with amlitelimab, in a Phase IIb trial over 52 weeks compared with placebo in patients with SSc-ILD (CONQUEST, NCT06195072) [44]. The B-cell-targeted mAb belimumab, currently approved for the treatment of LN and SLE, is being evaluated in adults with SSc-ILD (NCT05878717) [48], while the anti-type 1 IFN receptor antibody anifrolumab is being evaluated in a randomized Phase III trial in SSc (DAISY, NCT05925803). Patients are being stratified based on the presence or absence of ILD at baseline, with changes from baseline in FVC in patients with SSc-ILD as a key secondary end point [49].

Overall, this analysis showed that changes in lung function and skin fibrosis are predictive of mortality in SSc-ILD. Therefore, stabilizing these parameters might be the first step in a treat-to-target strategy before approaches leading to improvement become available. These findings are important as they demonstrate the link between short-term changes and mortality, independent of treatment, and will be applicable as new treatments are introduced.

The main strengths are the number of participants and the use of both clinical and physiological variables. Limitations include the *post hoc* nature of the analyses, precluding standardization in the selection, initial evaluation and follow-up visits of patients. Indeed, several patients had no follow-up visit within 12 months after the baseline visit or after the 12 months' evaluation, as expected for an observational registry study. Also, all patients had radiologically confirmed ILD, but not all had HRCT data available, and the date of confirmed ILD diagnosis was missing for 47% of patients with prevalent ILD. In these patients, the onset of non-RP symptoms was used as an indicator of disease progression. This likely had minimal impact on the results, as half of the patients with a recorded ILD onset date had a confirmed ILD diagnosis for over 3 years. Additionally, previous EUSTAR data indicate that approximately one-quarter of patients experience progressive ILD, defined as an FVC decline of $\geq 5\%$, in each 12-month period over a 5-year follow-up, suggesting a relatively stable progression rate [32]. Finally, exhaustive haemodynamic data are not available for large cohorts as right heart catheterization is usually performed only in case of suspected PH. In the multivariable sensitivity analysis, which included the variable PH suspected by echocardiogram in the model, results were similar for the different predictors, although not statistically significant. This can be explained

by a lower number of cases included in the analysis (726 patients out of 893 had available data) or collinearity between variables (DL_{CO} and presence of PH or ILD, for example). When patients with suspected PH were excluded, HRs for mortality were stable.

These data confirm that a decline in FVC or a combined FVC/DL_{CO} or FVC/mRSS deterioration are predictive for mortality and may be particularly valuable for interpreting clinical trials in SSc-ILD. These data give insights into how changes in these parameters may affect mortality. These findings also support the regular assessment of decline in FVC% pred, which should be taken into consideration when evaluating risk of mortality.

Supplementary material

Supplementary material is available at Rheumatology online.

Data availability

Data are available upon reasonable request.

Funding

This study was partially funded by Boehringer Ingelheim International GmbH. The authors meet criteria for authorship as recommended by the International Committee of Medical Journal Editors (ICMJE). The authors received no direct compensation related to the development of the manuscript. This analysis, using the EUSTAR database, was funded by Boehringer Ingelheim. Boehringer Ingelheim was given the opportunity to review the article for medical as well as scientific accuracy and intellectual property considerations, if applicable. Authors maintained full editorial control over the content of the article and were responsible for all final decisions on article content, for final approval of the version for submission and of the version for publication.

Disclosure statement: Vincent Sobanski has received congress/travel funding from Boehringer Ingelheim and reports consultancies and speaking fees from Fresenius Kabi and Grifols and research support from Actelion, Grifols, GlaxoSmithKline, Octapharma, Pfizer and Shire, outside the submitted work. Jeska de Vries-Bouwstra has received grant/ research support to her institution from ReumaNederland (Dutch patient society for rheumatology), Nationale Vereniging voor mensen met lupus, APS, sclerodermie en MCTD (Dutch patient society), ARCH (Autoimmune Research and Collaboration Hub; Dutch interdisciplinary society for patients and caregivers), Roche, Galapagos, Janssen-Cilag and Boehringer Ingelheim; consulting fees from Boehringer Ingelheim, Janssen-Cilag and AbbVie (payments made to institution); speaker fees from Dutch Society of Rheumatology (payments made to institution), Boehringer Ingelheim (payments made to institution) and Janssen-Cilag (payments made to institution). Anna-Maria Hoffmann-Vold has received speaker fees from Boehringer Ingelheim, Janssen, Medscape, Merck Sharp & Dohme, Novartis and Roche; consultancy fees from AbbVie, ARXX, BMS, Boehringer Ingelheim, Genentech, Janssen, Medscape, Merck Sharp & Dohme, Pliant Therapeutics, Roche and Werfen; and research/grant support from Boehringer Ingelheim and Janssen. She is also group leader for the European Alliance of Associations for Rheumatology study group on the lung in rheumatic and musculoskeletal diseases. Margarida Alves is an employee of Boehringer Ingelheim. Marco Matucci-Cerinic has received personal fees from Boehringer Ingelheim. László Czirják has received honoraria for consulting services (educational services and participation in advisory boards) from Boehringer Ingelheim, Roche, Merck Sharp 87. Dohme, Pfizer, AbbVie, UCB, Lilly, GlaxoSmithKline, Novartis, Bayer and Richter. Otvlia Kowal-Bielecka has received consultancy fees and speakers' bureau and/or congress/travel funding from AbbVie, Bayer, CSL Behring, Boehringer Ingelheim, Gilead, Medac, Merck Sharp & Dohme, Novartis, Pfizer, Roche and Sandoz. Yannick Allanore had consultancy relationships or received research grants from Argenx, AstraZeneca, Bayer, Alpine Immuno-Sciences, Boehringer Ingelheim, Galderma, Genentech/Roche, Janssen, Horizon, Medsenic, Merck Serono, Prometheus and Topadur. Nils Schoof was previously an employee of Boehringer Ingelheim. Oliver Distler has/had a consultancy relationship with and/or has received research funding from and/or has served as a speaker for the following companies in the area of potential treatments for systemic sclerosis and its complications in the last three calendar years: 4P-Pharma, AbbVie, Acceleron, Alcimed, Altavant, Amgen, AnaMar, Argenx, Arxx, AstraZeneca, Blade, Bayer, Boehringer Ingelheim, Cantargia AB, Corbus, CSL Behring, Galderma, Galapagos, Glenmark, Gossamer, Horizon, Janssen, Kymera, Lupin, Medscape, Merck, Miltenvi Biotec, Mitsubishi Tanabe, Nkarta Inc., Novartis, Orion, Prometheus, Redxpharma, Roivant, EMD Serono, Topadur and UCB. He has a patent issued: 'mir-29 for the treatment of systemic sclerosis' (US8247389, EP2331143) and is co-founder of CITUS AG. Dörte Huscher, Gabriela Riemekasten and Mengtao Li have nothing to declare.

Acknowledgements

The authors would like to acknowledge the valuable contributions of Dr Gabriele Valentini in the development of this article. Editorial support was provided by John Carron, PhD, and Marketa Borkovcova, MSc, of Nucleus Global, which was contracted and funded by Boehringer Ingelheim. A list of the EUSTAR collaborators can be found in Supplementary Data S2, available at *Rheumatology* online.

References

- 1. Denton CP, Khanna D. Systemic sclerosis. Lancet 2017;390: 1685–99. doi: 10.1016/S0140-6736(17)30933-9
- Steen VD, Medsger TA. Changes in causes of death in systemic sclerosis, 1972–2002. Ann Rheum Dis 2007;66:940–4. doi: 10.1136/ ard.2006.066068
- Elhai M, Meune C, Boubaya M et al.; EUSTAR group. Mapping and predicting mortality from systemic sclerosis. Ann Rheum Dis 2017;76:1897–905. doi: 10.1136/annrheumdis-2017-211448
- Hoffmann-Vold AM, Fretheim H, Halse AK *et al.* Tracking impact of interstitial lung disease in systemic sclerosis in a complete nationwide cohort. Am J Respir Crit Care Med 2019;200:1258–66. doi: 10.1164/rccm.201903-0486OC
- Volkmann ER. Natural history of systemic sclerosis-related interstitial lung disease: how to identify a progressive fibrosing phenotype. J Scleroderma Relat Disord 2020;5:31–40. doi: 10.1177/ 2397198319889549

- Maher TM, Bourdin A, Volkmann ER *et al.* Decline in forced vital capacity in subjects with systemic sclerosis-associated interstitial lung disease in the SENSCIS trial compared with healthy reference subjects. Respir Res 2022;23:178. doi: 10.1186/s12931-022-02095-6
- Distler O, Assassi S, Cottin V et al. Predictors of progression in systemic sclerosis patients with interstitial lung disease. Eur Respir J 2020;55:1902026. doi: 10.1183/13993003.02026-2019
- Winstone TA, Assayag D, Wilcox PG et al. Predictors of mortality and progression in scleroderma-associated interstitial lung disease: a systematic review. Chest 2014;146:422–36. doi: 10.1378/chest. 13-2626
- Goh NS, Desai SR, Veeraraghavan S et al. Interstitial lung disease in systemic sclerosis: a simple staging system. Am J Respir Crit Care Med 2008;177:1248–54. doi: 10.1164/rccm.200706-877OC
- Morisset J, Vittinghoff E, Elicker BM *et al.* Mortality risk prediction in scleroderma-related interstitial lung disease: the SADL model. Chest 2017;152:999–1007. doi: 10.1016/j.chest.2017.06.009
- Ebata S, Yoshizaki A, Oba K *et al.* Safety and efficacy of rituximab in systemic sclerosis (DESIRES): open-label extension of a doubleblind, investigators-initiated, randomised, placebo-controlled trial. Lancet Rheumatol 2022;4:e546–55. doi: 10.1016/s2665-99 13(22)00131-x
- Khanna D, Lin CJF, Furst DE *et al.*; focuSSced investigators. Tocilizumab in systemic sclerosis: a randomised, double-blind, placebo-controlled, phase 3 trial. Lancet Respir Med 2020;8: 963–74. doi: 10.1016/S2213-2600(20)30318-0
- Khanna D, Albera C, Fischer A *et al.* An open-label, phase II study of the safety and tolerability of pirfenidone in patients with scleroderma-associated interstitial lung disease: the LOTUSS trial. J Rheumatol 2016;43:1672–9. doi: 10.3899/jrheum.151322
- Perelas A, Silver RM, Arrossi AV, Highland KB. Systemic sclerosis-associated interstitial lung disease. Lancet Respir Med 2020;8:304–20. doi: 10.1016/S2213-2600(19)30480-1
- Guler SA, Winstone TA, Murphy D et al. Does systemic sclerosisassociated interstitial lung disease burn out? Specific phenotypes of disease progression. Ann Am Thorac Soc 2018;15:1427–33. doi: 10.1513/AnnalsATS.201806-362OC
- Volkmann ER, Tashkin DP, Sim M et al.; SLS I and SLS II study groups. Short-term progression of interstitial lung disease in systemic sclerosis predicts long-term survival in two independent clinical trial cohorts. Ann Rheum Dis 2019;78:122–30. doi: 10.1136/ annrheumdis-2018-213708
- Goh NS, Hoyles RK, Denton CP et al. Short-term pulmonary function trends are predictive of mortality in interstitial lung disease associated with systemic sclerosis. Arthritis Rheumatol 2017;69: 1670–8. doi: 10.1002/art.40130
- Mihai C, Landewe R, van der Heijde D et al.; EUSTAR co-authors. Digital ulcers predict a worse disease course in patients with systemic sclerosis. Ann Rheum Dis 2016;75:681–6. doi: 10.1136/ annrheumdis-2014-205897
- Pope JE. The future of treatment in systemic sclerosis: can we design better trials? Lancet Rheumatol 2020;2:e185–94. doi: 10.1016/S2665-9913(20)30010-2
- Meier FM, Frommer KW, Dinser R *et al.*; EUSTAR Co-authors. Update on the profile of the EUSTAR cohort: an analysis of the EULAR Scleroderma Trials and Research group database. Ann Rheum Dis 2012;71:1355–60. doi: 10.1136/annrheumdis-2011-200742
- 21. Masi AT, Rodnan GP, Medsger TA Jr, *et al.*; Subcommittee for Scleroderma Criteria of the American Rheumatism Association Diagnostic and Therapeutic Criteria Committee. Preliminary criteria for the classification of systemic sclerosis (scleroderma). Arthritis Rheum 1980;23:581–90. doi: 10.1002/art.1780230510
- 22. van den Hoogen F, Khanna D, Fransen J et al. 2013 classification criteria for systemic sclerosis: an American College of Rheumatology/ European League Against Rheumatism collaborative initiative. Arthritis Rheum 2013;65:2737–47. doi: 10.1002/art.38098

- 23. Sobanski V, Giovannelli J, Allanore Y et al.; EUSTAR Collaborators. Phenotypes determined by cluster analysis and their survival in the prospective European Scleroderma Trials and Research cohort of patients with systemic sclerosis. Arthritis Rheumatol 2019;71:1553–70. doi: 10.1002/art.40906
- Distler O, Brown KK, Distler JHW et al.; SENSCISTM trial investigators. Design of a randomised, placebo-controlled clinical trial of nintedanib in patients with systemic sclerosis-associated interstitial lung disease (SENSCIS). Clin Exp Rheumatol 2017;35:75–81.
- 25. Maurer B, Graf N, Michel BA et al.; EUSTAR co-authors. Prediction of worsening of skin fibrosis in patients with diffuse cutaneous systemic sclerosis using the EUSTAR database. Ann Rheum Dis 2015;74:1124–31. doi: 10.1136/annrheumdis-2014-205226
- Showalter K, Hoffmann A, Rouleau G *et al.* Performance of forced vital capacity and lung diffusion cutpoints for associated radiographic interstitial lung disease in systemic sclerosis. J Rheumatol 2018;45:1572–6. doi: 10.3899/jrheum.171362
- 27. Castelino FV, Dellaripa PF. Recent progress in systemic sclerosisinterstitial lung disease. Curr Opin Rheumatol 2018;30:570–5. doi: 10.1097/BOR.00000000000544
- Caron M, Hoa S, Hudson M, Schwartzman K, Steele R. Pulmonary function tests as outcomes for systemic sclerosis interstitial lung disease. Eur Respir Rev 2018;27:170102. doi: 10.1183/16000617.0102-2017
- Petelytska L, Bonomi F, Cannistra C et al. Heterogeneity of determining disease severity, clinical course and outcomes in systemic sclerosisassociated interstitial lung disease: a systematic literature review. RMD Open 2023;9:e003426. doi: 10.1136/rmdopen-2023-003426
- Kafaja S, Clements PJ, Wilhalme H et al. Reliability and minimal clinically important differences of forced vital capacity: results from the Scleroderma Lung Studies (SLS-I and SLS-II). Am J Respir Crit Care Med 2018;197:644–52. doi: 10.1164/rccm.201709-1845OC
- Khanna D, Mittoo S, Aggarwal R *et al.* Connective tissue diseaseassociated interstitial lung diseases (CTD-ILD) – report from OMERACT CTD-ILD Working Group. J Rheumatol 2015;42: 2168–71. doi: 10.3899/jrheum.141182
- 32. Hoffmann-Vold A-M, Allanore Y, Alves M *et al.*; EUSTAR collaborators. Progressive interstitial lung disease in patients with systemic sclerosis-associated interstitial lung disease in the EUSTAR database. Ann Rheum Dis 2021;80:219–27. doi: 10.1136/annrheumdis-2020-217455
- Le Gouellec N, Duhamel A, Perez T et al. Predictors of lung function test severity and outcome in systemic sclerosis-associated interstitial lung disease. PLoS One. 2017;12:e0181692. doi: 10.1371/journal.pone.0181692
- Steen VD, Conte C, Owens GR, Medsger TA Jr. Severe restrictive lung disease in systemic sclerosis. Arthritis Rheum 1994;37: 1283–9. doi: 10.1002/art.1780370903
- Moore OA, Proudman SM, Goh N *et al.* Quantifying change in pulmonary function as a prognostic marker in systemic sclerosisrelated interstitial lung disease. Clin Exp Rheumatol. 2015; 33:S111–6.
- Hoffmann-Vold A-M, Maher TM, Philpot EE *et al.* The identification and management of interstitial lung disease in systemic sclerosis: evidence-based European consensus statements. Lancet Rheumatol 2020;2:e71–83. doi: 10.1016/S2665-9913(19)30144-4
- Ryerson CJ, Vittinghoff E, Ley B *et al.* Predicting survival across chronic interstitial lung disease: the ILD-GAP model. Chest 2014; 145:723–8. doi: 10.1378/chest.13-1474
- Wu W, Jordan S, Becker MO *et al.* Prediction of progression of interstitial lung disease in patients with systemic sclerosis: the SPAR model. Ann Rheum Dis 2018;77:1326–32. doi: 10.1136/annrheumdis-2018-213201
- 39. Wu W, Jordan S, Graf N et al.; EUSTAR Collaborators. Progressive skin fibrosis is associated with a decline in lung function and worse survival in patients with diffuse cutaneous systemic sclerosis in the European Scleroderma Trials and Research (EUSTAR) cohort. Ann Rheum Dis 2019;78:648–56. doi: 10.1136/annrheumdis-2018-213455

- Raghu G, Montesi SB, Silver RM *et al.* Treatment of systemic sclerosis-associated interstitial lung disease: evidence-based recommendations. An official American Thoracic Society Clinical Practice Guideline. Am J Respir Crit Care Med 2024;209:137–52. doi: 10.1164/rccm.202306-1113ST
- 41. Campochiaro C, Lazzaroni MG, Bruni C *et al.* Open questions on the management of targeted therapies for the treatment of systemic sclerosis-interstitial lung disease: results of a EUSTAR survey based on a systemic literature review. Ther Adv Musculoskelet Dis 2022; 14:1759720X221116408. doi: 10.1177/1759720X221116408
- Distler O, Highland KB, Gahlemann M et al.; SENSCIS Trial Investigators. Nintedanib for systemic sclerosis-associated interstitial lung disease. N Engl J Med 2019;380:2518–28. doi: 10.1056/NEJMoa1903076
- Ebata S, Yoshizaki-Ogawa A, Sato S, Yoshizaki A. New era in systemic sclerosis treatment: recently approved therapeutics. J Clin Med 2022;11:4631. doi: 10.3390/jcm11154631
- 44. Distler O, Allanore Y, Denton CP *et al.* Riociguat in patients with early diffuse cutaneous systemic sclerosis (RISE-SSc): open-label, long-term extension of a phase 2b, randomised, placebo-controlled trial. Lancet Rheumatol. 2023;5:e660–9. doi: 10.1016/s2665-9913 (23)00238-2

- 45. Khanna D, Allanore Y, Denton CP *et al.* Riociguat in patients with early diffuse cutaneous systemic sclerosis (RISE-SSc): randomised, double-blind, placebo-controlled multicentre trial. Ann Rheum Dis 2020;79:618–25. doi: 10.1136/annrheumdis-2019-216823
- 46. Herrmann FE, Hesslinger C, Wollin L, Nickolaus P. BI 1015550 is a PDE4B inhibitor and a clinical drug candidate for the oral treatment of idiopathic pulmonary fibrosis. Front Pharmacol 2022;13: 838449. doi: 10.3389/fphar.2022.838449
- Richeldi L, Azuma A, Cottin V *et al.*; 1305-0013 Trial Investigators. Trial of a preferential phosphodiesterase 4B inhibitor for idiopathic pulmonary fibrosis. N Engl J Med 2022;386: 2178–87. doi: 10.1056/NEJMoa2201737
- Ferrari HM, Kale-Pradhan P, Konja J, Dierker M, Martirosov AL. Systemic-sclerosis-related interstitial lung disease: a review of the literature and recommended approach for clinical pharmacists. Ann Pharmacother 2024;58:956–71. doi: 10.1177/10600280231213672
- 49. Khanna D, Denton CP, Assassi S et al. A randomised, parallelgroup, double-blind, placebo-controlled phase 3 study to Determine the effectiveness of the type I interferon receptor antibody, Anifrolumab, In SYstemic sclerosis: DAISY study design and rationale. Clin Exp Rheumatol 2024;42:1635–44. doi: 10.55563/clinexprheumatol/s8qcyu

© The Author(s) 2025. Published by Oxford University Press on behalf of the British Society for Rheumatology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited. Rheumatology, 2025, 00, 1–10

https://doi.org/10.1093/rheumatology/keaf264 Original Article